Author's response to reviews

Title: A case of phacolytic glaucoma with anterior lens capsule disruption identified by scanning electron microscopy

Authors:

Woong-Sun Yoo (oocee@naver.com)
Byeong-Jae Kim (nadadada@hanmail.net)
In-Young Chung (in0chung@hanmail.net)
Seong-Wook Seo (stramast@naver.com)
Ji-Myong Yoo (yjm@gnu.ac.kr)
Seong-Jae Kim (oocee@hanmail.net)

Version: 4 Date: 29 August 2014

Author's response to reviews: see over
August, 14, 2014

Alice Murray  
Editor-in-Chief  
*BioMed Central*

Dear Dr. Murray

Please find attached the manuscript entitled “A case of phacolytic glaucoma with anterior lens capsule disruption identified by scanning electron microscopy” for consideration for publication in *BMC Ophthalmology*.

This manuscript describes the anterior capsule lens in a phacolytic glaucoma patient which showed multiple disruptions with scanning electron microscopy while it revealed intact appearance with light microscopy. This is the first report that confirming the lens anterior capsule disruptions in phacolytic glaucoma. We feel that this results will be of interest to the journal readers.

This manuscript is an original work and has been seen and approved by all of the authors. I had full access to all the data in the study and take responsibility for the integrity of the data and the accuracy of the data analysis as well as the decision to submit for publication. This manuscript is not currently being considered by another journal for publication. Thank you for considering our manuscript for publication, and I look forward to hearing from you.

Sincerely,

Seong-Jae, Kim  
Department of Ophthalmology  
School of Medicine, Institute of Health Science, Gyeongsang National University Hospital  
79 Gangnam-ro, Jinju  
Gyeongnam 660-702  
South Korea  
E-mail: maya12kim@naver.com  
Tel: +82-55-750-8468  
Fax: +82-55-758-4158